Author’s response to reviews

Title: A mechanistic approach to anti-nociceptive potential of Artemisia macrocephala Jacquem.

Authors:
Mohammad Shoaib (mohammadshoaib13@yahoo.com)
Ismail Shah (ismailshah09@yahoo.com)
Niaz Ali (niazpharmacist@yahoo.com)
Syed Shah (pharmacistsyed@gmail.com)

Version: 4 Date: 28 Feb 2016

Author’s response to reviews:

Reviewer Comments for Manuscript BCAM-D-15-00072R3 "A mechanistic approach to anti-nociceptive potential of Artemisia macrocephala Jacquem."

Reviewer #3: The authors did not make the corrections that were requested previously. They did not provide answer for corrections such as performing a chemical characterization of the plant fractions.

Rectifications of reviewers’ comments:

The fractions were chemically characterized for the presence of various phytochemicals like flavonoids, terpenes, saponins and also the bio-active fractions were subjected to column chromatography for isolation of compounds. The detail structural data and also the spectral data are provided in manuscript/ supporting documents.

We appreciate the efforts and comments of reviewers to make this manuscript of standard.

Hope you will give now a prompt decision.

Regards!

Ismail Shah.